Author’s response to reviews

Title: Predominant Global Glomerulosclerosis in Patients of Upper Urinary Tract Urothelial Carcinoma with Pre-existing Renal Function Impairment is a Predictor of Poor Renal Outcomes

Authors:

Sheng-Wen Niu (950138kmuh@gmail.com)
Peir-In Liang (peirinl@yahoo.com)
Ming-Yen Lin (mingyenlin3@gmail.com)
Shih-Meng Yeh (yesmall83@yahoo.com.tw)
Yen-Yi Zhen (yenyizhen@gmail.com)
Yu-Han Chang (1038043@kmth.org.tw)
Pin-Chia Huang (1028264@kmuh.org.tw)
Chi-Chi Hung (chichi@kmu.edu.tw)
I-Ching Kuo (980135kmuh@gmail.com)
Hugo Lin (yukenlin@yahoo.com.tw)
Mei-Chuan Kuo (mechku@cc.kmu.edu.tw)
Wei-Ming Li (u8401067@yahoo.com.tw)
Chun-Nung Huang (cnhuang.uro@gmail.com)
Wen-Jeng Wu (wejewu@cc.kmu.edu.tw)
Li-Tzong Chen (leochen@nhri.org.tw)
Yi-Wen Chiu (chiuyiwen@gmail.com)
Shang-Jyh Hwang (sjhwang@kmu.edu.tw)

Version: 3 Date: 18 Feb 2019

Author’s response to reviews:

Q1: Please remove your 'Word Count' section from your title page.
Ans1: Yes, we have made this revision by reviewer’s suggestion on title page.

Q2: If the need for consent was waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this in the 'Ethics approval and consent to participate' section, including the name of the IRB or a reference to the relevant legislation.

Ans2: Yes, we have made this revision by reviewer’s suggestion in 'Ethics approval and consent to participate' section (Line 1-2, page 22). The study protocol was approved by the institutional review board (IRB) of Kaohsiung medical university (Kaohsiung, Taiwan). The need for consent was waived by the IRB (KMUH-IRB-20120138) and the data was decoded before undergoing analysis. All demographic data and histopathologic data were collected and fulfilled the regulations of this IRB of our hospital.

Q3: Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

Ans3: We appreciate your comments. We have made these revisions by reviewer’s suggestion in 'Figure Legends' section (Line1-7, page 33).

Q4: Multi-panel figures (those with parts a, b, c, d etc.) should be submitted as a single composite file that contains all parts of the figure.

Ans4: Yes, we have made this revision by reviewer’s suggestion for our figures.

Q5: Thank you for providing a cover letter. As this document is no longer required at this stage of the publication process, please remove it from your submission’s supplementary files.

Ans5: Yes, we have removed the cover letter.

Q6: At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Ans6: We appreciate your comments.